Author's response to reviews

Title: Development of a Short Form and Scoring Algorithm from the Validated Actionable Multiple Sclerosis Bladder Health Screening Tool

Authors:

David Bates (David.Bates@nuth.nhs.uk)
Jack Burks (ack@jackburks.com)
Denise Globe (Globe_Denise@Allergan.com)
Manuel Signori (Signori_Manuel@Allergan.com)
Stacie Hudgens (stacie.hudgens@adelphivalues.com)
Pierre Denys (denysaphp@gmail.com)
Scott MacDiarmid (smacdiarmid@allianceurology.com)
Victor Nitti (Victor.Nitti@nyumc.org)
Ib Odderson (odderson@u.washington.edu)
Amy Perrin Ross (aperrinros@aol.com)
Michael Chancellor (Michael.chancellor@beaumont.edu)

Version: 2 Date: 17 October 2012

Author's response to reviews: see over
(1) Ethics Statement:

Updated to include the name of the our IRB (Copernicus Group) as well as our protocol number (MAP1-11-049) for this project.

(2) Consent Statement:

Added “Written informed consent was needed before the subject could participate in any part of the study”.

(3) Acknowledgments:

Added the acknowledgments section.